Author’s response to reviews

Title: Intensified household contact tracing, prevention and treatment support versus enhanced standard of care for contacts of tuberculosis cases in South Africa: Study protocol for a household cluster-randomised trial

Authors:

Peter MacPherson (peter.macpherson@lstmed.ac.uk)
Emily Webb (emily.webb@lshtm.ac.uk)
Ebrahim Variava (variava@wordlonline.co.za)
Sanjay Lala (sanjay.lala@wits.ac.za)
Minja Milovanovic (milovanovicm@phru.co.za)
Andrew Ratsela (Rampule.Ratsela@ul.ac.za)
Limakatso Lebina (lebinal@phru.co.za)
Anthony Kinghorn (kinghorna@phru.co.za)
Neil Martinson (martinson@phru.co.za)

Version: 1 Date: 19 Sep 2019

Author’s response to reviews:

Dear Eva Szunyogova,

Re: INFD-D-19-00112. Intensified household contact tracing, prevention and treatment support versus enhanced standard of care for contacts of tuberculosis cases in South Africa: a household cluster-randomised trial

Many thanks. We have added references to the two supplemental files (map of study districts and SPIRIT Checklist) within manuscript as requested.

Kind regards,

Peter MacPherson